

# **ORCA - Online Research @ Cardiff**

This is an Open Access document downloaded from ORCA, Cardiff University's institutional repository:https://orca.cardiff.ac.uk/id/eprint/130726/

This is the author's version of a work that was submitted to / accepted for publication.

Citation for final published version:

Cannas, Violetta, Gosling, Jonathan, Pero, Margherita and Rossi, Tommaso 2020. Determinants for order-fulfilment strategies in engineer-to-order companies: Insights from the machinery industry. International Journal of Production Economics 228, 107743. 10.1016/j.ijpe.2020.107743

Publishers page: https://doi.org/10.1016/j.ijpe.2020.107743

#### Please note:

Changes made as a result of publishing processes such as copy-editing, formatting and page numbers may not be reflected in this version. For the definitive version of this publication, please refer to the published source. You are advised to consult the publisher's version if you wish to cite this paper.

This version is being made available in accordance with publisher policies. See http://orca.cf.ac.uk/policies.html for usage policies. Copyright and moral rights for publications made available in ORCA are retained by the copyright holders.



# Determinants for Order Fulfilment Strategies in Engineer-to-Order companies: insights from the Machinery Industry

#### **Abstract**

Recent empirical studies have refined our understanding of Engineer-to-order (ETO) situations, supporting the existence of different order fulfilment strategies based on the degree of customer involvement in the engineering and production activities, which differs depending on the strategic fit with the environment where the company operates. Despite the importance of this finding, limited attempts have been made to provide a comprehensive understanding of the determinants for this strategic choice in ETO companies. To overcome this gap, this study aims at investigating what are the sources of differentiation between the environments that ETO companies can face and how they react to strategically fit the order fulfilment strategy. In doing so, this research analyses the existing literature through a contingency theory lens and performs a multiple case study research in a specific ETO sector, i.e., the machinery industry. The study identifies five different order fulfilment strategies implemented in the machinery industry to provide different product families to the market. For each of them, the different environment characteristics have been defined and the performance outcome has been measured, explaining the rationale for the positioning of the product families in different strategies. The findings of this paper suggest two main contributions. First, the study contributes to theory, by deepening and refining the analysis of contingencies for choosing different order fulfilment strategies in the ETO context. Second, the study provides practical guidelines to ETO companies that want to adapt their order fulfilment strategy to the unexpected or planned changes in their environment.

Keywords. Engineer-to-order, Machinery Industry, Order fulfilment strategy, Customer order decoupling point, Case study research

#### 1 Introduction

The order fulfilment strategy defines the way that a company responds to a customer order, i.e., the activities performed from the sales inquiry to the delivery of the product (Kritchanchai and MacCarthy, 1999; Lambert and Cooper, 2000; Shapiro et al., 1992). Thus, a strategic choice is related to the location of the customer order decoupling point (CODP), which separates forecast-driven and order-driven activities (Hoekstra and Romme, 1992; Sharman, 1984; Wemmerlöv, 1984; Wortmann, 1983). A range of possible order fulfilment strategies can be identified based on the location of the CODP (Lampel and Mintzberg, 1996): from pure customised Engineer-to-Order (ETO), where products are designed and produced according to customer specifications, to pure standardised Make-to-Stock (MTS), where products are delivered from finished goods (Wortmann et al., 1997).

Engineer-to-order (ETO) companies can choose among different order fulfilment strategies. Literature suggests that ETO companies can perform different activities after the customer order, e.g. modify existing designs or develop an entirely new one for every customer order (Gosling and Naim, 2009). Typically, ETO companies perform a certain amount of engineering and manufacturing activities both before and after the order arrival (Adrodegari et al., 2015). The traditional understanding of ETO situations, i.e., companies designing and producing each product from scratch following customer specifications (Caron and Fiore, 1995), has been recently refined to a more comprehensive understanding (Gosling et al., 2017; Willner et al., 2016). Different archetypes have been identified depending on the degree of design standardisation and the consequent amount of engineering and production work performed after the order.

When dealing with the order fulfilment strategies in ETO companies, then, the traditional conceptualisation of the CODP is problematic. Seminal CODP studies (Hoekstra and Romme, 1992; Sharman, 1984; Wortmann, 1983) consider engineering as a part of a linear sequence of activities that can be performed before or after CODP. However, to remain competitive, ETO companies accommodate different order fulfilment strategies, thus facing the particular challenge of having to integrate and manage both order-driven and forecast-driven engineering and production activities (Cannas et al., 2019). Hence, it is more useful to follow a two-dimensional approach for the CODP positioning, by defining both the amount of engineering and production work to perform before and after the order (Dekkers, 2006; Wikner and Rudberg, 2005).

The choice of the order fulfilment strategy represents a significant decision for ETO companies, given its serious implications for performance in terms of lead time, price, flexibility and quality (Olhager, 2003). This is even more relevant in the recent competitive arena, where ETO companies are claimed to offer high product variety in short time (André and Elgh, 2018; Cannas et al., 2018; Haug et al., 2009; Kristjansdottir et al., 2018; Sylla et al., 2018). High customisation can increase costs and lead times. As ETO companies leverage on a range of different product customisation choices, some of them highly customised while others can be more standardised (Hicks et al., 2000; McGovern et al., 1999). Furthermore, the implications for standardisation and efficiency in ETO companies of anticipating engineering activities have been underlined also by very recent studies that analyse more in-depth the application of lean, modularity and product platforms techniques (André and Elgh, 2018; Birkie and Trucco, 2016; Johansson and Elgh, 2019; Johnsen and Hvam, 2018).

Many complex engineering projects are plagued by delays and cost overruns (Flyvbjerg, 2014). Practical examples can be found in the media concerning construction, aerospace, capital goods and shipbuilding industries; e.g., the project for the International Space Station

was completed six years late and 186% over-budget (Forbes, 2018); Boeing 787 Dreamliner was billions of dollars over budget and three years behind schedule (Forbes, 2013). Despite the above, literature is falling short in providing adequate support to ETO companies in choosing the appropriate order fulfilment strategy (Sandrin et al., 2018).

The CODP positioning along the continuum of activities of the production process has been demonstrated to be contingent upon a set of external and internal factors: market, product and production process characteristics (Van Donk, 2001). Research that has addressed the classification of ETO companies based on the extent to which they exploit existing designs or explore new knowledge, demonstrates that it depends on the environment to which they have to adapt (Cannas et al., 2019; Dekkers, 2006; Gosling et al., 2017; Semini et al., 2014; Wikner and Rudberg, 2005; Willner et al., 2016). These studies have made significant efforts to integrate the engineering process in the CODP concept by empirically analysing the application of this concept to complex ETO realities. However, none of these studies provides a comprehensive and specific framework related to the determinants for the CODP positioning in both the engineering and the production processes.

Consequently, it is difficult to provide practical guidelines to support ETO companies in choosing the appropriate order fulfilment strategy. For example, should a producer of ETO products aim to develop a fulfilment strategy that delivers customised products reusing existing designs, or to create more standard offerings by pursuing modular principles? The response to this question could depend on the environment the ETO company operates in and would be helpful to understand whether the current order fulfilment strategy supports the achievement of the desired performance outcome (Haug et al., 2009; Kristianto et al., 2013; Schoenwitz et al., 2017).

Hence, this study aims at understanding the sources of differentiation between the environments that ETO companies can face and how ETO companies react to strategically fit the order fulfilment strategy to the environment. For this reason, this article seeks to explore and explain this phenomenon, by answering the following research questions:

RQ1. What are the determinants for the order fulfilment strategy choices within ETO companies?

RQ2. How can ETO companies fit their order fulfilment strategies to different environments? To achieve this aim, the researchers first reviewed the previous literature related to the CODP positioning through a contingency theory lens. Based on the review of literature, the initial framework of the study was developed, and the current state-of-the-art and research gaps were identified. Then, a multiple case study research was conducted in a specific ETO sector,

i.e., the machinery industry. The remainder of this paper is organized as follows. Section 2 presents the literature review and section 3 defines the research methodology. Section 4 and section 5 present the results of the case studies. Section 6 discusses the results and section 7 provides the conclusion of the research, presenting the main implications and limitations of the study.

#### 2 Literature review

#### 2.1 Order fulfilment strategies in ETO companies

The first studies related to CODP based order fulfilment strategies dated back to the 80s and the 90s (Hoekstra and Romme, 1992; Sharman, 1984; Wemmerlöv, 1984; Wortmann, 1983). In this period, the increasing demand for innovative products triggered the movement towards more customised manufacturing systems (Wortmann, 1992) with the beginning of a new industrial paradigm, i.e., mass customisation (Davis, 1989; Pine, 1993). Accordingly, a continuum of order fulfilment strategies was demonstrated to exist based on the point of customer involvement (Lampel and Mintzberg, 1996), i.e., the CODP location, from pure standardisation (i.e., MTS), to pure customisation (i.e., ETO): including segmented standardisation, i.e., delivery-to-order (DTO), customised standardisation, i.e., assembly-to-order (ATO), and tailored customisation, i.e., make-to-order (MTO). Since these seminal studies, the CODP positioning has been introduced as a strategic choice.

Further studies included engineering activities into the frameworks of CODP. Among the others, Amaro et al. (1999) included in their taxonomy of order fulfilment strategies the degree of design customisation, in line with Mintzberg (1988), from "pure" (engineer completely new designs) to "tailored" (adapt existing designs), "standardised" (combine existing options) and "none" (take the design as it is). They stated that there are no reasons to assume a one-to-one correlation between the activities performed after the customer order and the degree of design customisation. Thus, they increased the number of order fulfilment strategies according to the different potential combinations of these two dimensions. Similarly, a range of four different ideal ETO types has been proposed by Hicks et al. (2001), which analysed the new phenomenon of more standardised modular designs.

These studies were considered not sufficient to cover the gap between theory and practice existing in the ETO literature. Indeed, Rahim and Baksh (2003), analysing the differences between ETO and MTS, affirmed that the literature is too much focused on MTS, whereas ETO requires ad-hoc frameworks that show a structured approach to manage both engineering

and production processes and include the possibility of concurrency between the engineering and production activities.

To fill this gap, as underlined by Gosling and Naim (2009) and Dekkers et al. (2013), the two-dimensional (2D) CODP concept was introduced in the literature (Dekkers, 2006; Wikner and Rudberg, 2005), which extended the applicability of the CODP concept to a more comprehensive view. The 2D-CODP defines a broader order fulfilment strategy, where companies choose: (i) the number of engineering activities to perform to order; (ii) the number of production activities to perform to order. Thus, it decouples also the engineering process and takes into account the possibility of hybrid and overlapped strategies where engineering adaptations of existing designs are performed to order, in concurrency with production activities.

Since the 2D-CODP seminal studies, some attempts have been made in the literature over the years to classify the order fulfilment strategies by empirically analysing ETO companies. Veldman and Alblas (2012) analysed different engineering fulfilment strategies employed by two ETO companies, based upon a specific technology, predefined sub-functions and solution principles, and predefined finished goods. Semini et al. (2014) empirically analysed different engineering customisation strategies to design shipbuilding: (i) customised design, where most of the engineering activities are performed to order; (ii) and standardised design, where most of the engineering activities are performed to forecast, based on standardisation and modularisation. Additionally, different engineering standardisation and automation strategies have been identified by Willner et al. (2016) based on the engineering complexity and the production volumes, and different products of seven companies have been classified within them. The application of the decoupling concept to the engineering process has been further developed by Gosling et al. (2017), which investigated eight cases in different ETO projects. This study shows that the engineering decoupling choice covers a continuum of nine potential locations. Recently, based on the review of all the different existing studies abovementioned, Cannas et al. (2019) categorised different order fulfilment strategies in a structured 2D-CODP framework, which has been validated through the empirical insights of a multiple case study research in the machinery industry. This study shows that the focus of the managerial approaches to manage and coordinate engineering and production processes, before and after the decoupling point, varies depending on the decoupling strategy chosen.

# 2.2 Linking order fulfilment strategies and performance outcomes

The order fulfilment strategy aims to satisfy customer expectations in terms of flexibility, time, price and quality performance (Olhager, 2003). However, these different performance outcomes imply a trade-off: the achievement of high flexibility increases costs and lead times, and vice versa (Barlow et al., 2003). Consequently, the literature suggests that an operation cannot excel simultaneously on all performance measures and a company must define what key performance will be used for its success and focus on it (Chase et al., 2001). Therefore, the order fulfilment strategy should be implemented against the criteria that are important to the customer (Hallgren and Olhager, 2006; Olhager, 2010).

In the traditional connotation, the CODP positioning is considered to be dependent on the balance between: (i) competitive pressure, which pushes companies through forward shifting to improve delivery and price performance; and (ii) product cost (inventory management cost) and design complexity (customisation options offered to the market), which push companies through backwards CODP shifting to increase the customer service level (Hoekstra and Romme, 1992; Olhager, 2003; Sharman, 1984). The literature suggests to find the optimal balance between performance and eliminate trade-offs through the application of intermediate positions in the engineering and production CODP frameworks (Rudberg and Wikner, 2004). The aim is to be "lean" before the customer order, i.e. efficiency as winning criterion, and "agile" after the order, i.e. effectiveness as the winning criterion, or hybrid "leagile", i.e. both efficiency and effectiveness as winning criteria, in case of intermediate CODP positioning choices (Aitken et al., 2002; Christopher and Towill, 2001; Mason-Jones et al., 2000; Naylor et al., 1999).

The recent literature demonstrates that this assumption can also be appropriate to the 2D-CODP perspective and demands further analysis (Wikner and Rudberg, 2005). Therefore, the following studies analysed this topic and demonstrated that design reutilisation and standardisation in engineering work, i.e., forward shifting of the engineering CODP, reduces design costs and total lead time; whereas, design customisation, i.e., backwards shifting of the engineering CODP, increases the uniqueness and complexity of the design (i.e., innovativeness) as well as the lead times (Dekkers, 2006; Gosling et al., 2017). Recently, Cannas et al. (2019) empirically analysed the performance outcome of different order fulfilment strategies in ETO companies, demonstrating that they have an impact on five performance, including traditional measures of time, price, flexibility, and the innovativeness of the technology. They also included a new performance measure pertinent to the ETO sector,

i.e., reliability, which refers to the extent of risk for early unexpected defects after sales. In particular, this study proves that, when the engineering and production activities are mostly performed after the CODP, the performance outcome is the capability to always meet customer requirements for customisation and provide high technological innovation; vice versa, when the engineering and production activities are mostly performed before the CODP, competition is very high, and competitive prices are required, as well as short lead times and high reliability.

### 2.3 Aligning order fulfilment strategy to environment

When choosing the order fulfilment strategy, scholars agree that "one size does not fit all" (Jonsson and Rudberg, 2014; MacCarthy, 2013; Olhager, 2010; Schoenwitz et al., 2017; Van Donk and Van Doorne, 2016; Willner et al., 2016). According to the manufacturing strategy theory, the company should always align the market, product and processes choices to achieve good business performance, consistently with the degree of product customisation (Sousa and da Silveira, 2018)

In the logistics and manufacturing literature, the CODP positioning has been considered dependent on: (i) the relationship between the production lead time (P) and the delivery lead time (D), i.e., P/D ratio; and (ii) a set of contextual factors characterising the external and internal environment where the company operates (Hill, 1993; Hoekstra and Romme, 1992; Olhager and Wikner, 2000; Van Donk, 2001). A conceptual model was developed in the literature (Hallgren and Olhager, 2006; Olhager, 2003) to define all the factors affecting the CODP positioning and explain the motivations for backwards or forward shifting. The factors were allocated in three categories: market, product and production related factors. High demand volatility and low product volumes, as well as high product range and customisation requirements, are considered market characteristics that make impossible to provide products on an MTS basis. The production lead time is affected by the product structure (such as modularity, material profile, breadth and depth of the product structure) and the production process characteristics (such as the number of planning points, production process flexibility, and position of bottlenecks). This conceptual model represented an important basis to study the consequences of shifting the stock point through the materials flow (see for example: Hedenstierna and Ng, 2011; Liu et al., 2015; Sun et al., 2008).

In the supply chain management literature, Fisher (1997) was the first to introduce a model to match supply, product and demand and meet the needs of both the end consumers and the supply chains. In this study, the design of lean (i.e., physically efficient) or agile (market-responsive) supply chains was defined as matched respectively to functional (i.e., predictable

demand) and innovative (i.e., unpredictable demand) products. Following studies (Aitken et al., 2002; Christopher and Towill, 2001, 2002, 2000; Mason-Jones et al., 2000; Naylor et al., 1999) related the lean and agile supply chain concept to the CODP framework, associating the two paradigms to different order fulfilment strategies. The lean paradigm has been associated with the supply activities performed before the CODP, while the agile paradigm to the supply activities performed after the CODP. Accordingly, these works identified more than two distinguishing classes of supply chains and included the "leagile" concept, i.e., the possibility to combine lean and agile paradigms when companies apply hybrid standardisation and customisation strategies, such as the ATO one. In this sense, a taxonomy for defining the optimal strategy has been proposed by Cristopher et al. (2006). This study enriched the model previously proposed, analysing the supply chain design for global operations and defining three key factors affecting the strategic fulfilment choice: (i) supply characteristics (short/long replenishment lead time); (ii) product characteristics (special/standard); (iii) demand characteristics (predictable/unpredictable).

All these studies have contributed to support companies in facing the continuously changing market conditions and the increasing pressure on multiple competitive priorities. However, they present some limitations. The models proposed are focused mainly on the production dimension and the material flow, being very general about the engineering dimension and the design flow. Also, they do not consider any interfaces between engineering and production, looking at these processes as sequential. Therefore, these studies cannot be considered suitable to reflect the complexity that the ETO reality entails, where the products include different degrees of standardised and customised bill of materials (BoM) and engineering and production processes are strongly interconnected (Bozarth and Chapman, 1996; Giesberts and van der Tang, 1992; Hicks et al., 2000; Konijnendijk, 1993).

Some of the studies investigating the order fulfilment strategies in ETO companies suggest possible environmental factors affecting them. In particular, the environmental factors suggested to affect the order fulfilment strategies in ETO contexts can be grouped into three categories, namely: (i) market, (ii) product and (iii) process related factors (Hallgren and Olhager, 2006; Olhager, 2003). As far as market related factors are concerned, Schoenwitz et al. (2017) shown the importance of the alignment of customer preferences with the product design and process configuration. The market-related factors refer to the average annual units sold, market growth or general market changes, and customer requirements (Willner et al., 2016). As far as product related factors are concerned, Semini et al. (2014) proposed the product customisation/variety and modularity. Additionally, Willner et al. (2016) indicated the

engineering complexity as a determinant of customer order decoupling point positioning. As far as process related factors are concerned, with the addition of the engineering dimension, the P/D ratio has been demonstrated to be not sufficient anymore: the inclusion of the engineering lead time was needed (Wikner and Rudberg, 2005). In this case, the literature suggested that the major factor affecting the positioning is the relationship between the total lead time, which includes the engineering and production lead times, and the delivery lead time (Dekkers, 2006). In particular, Wikner and Rudberg (2005) considered the sum of engineering and production lead times as affected by a "delta value": null if they are sequential, or negative if they are overlapped. Moreover, Gosling et al. (2017) claimed that the engineering decoupling choice is contingent upon the engineering process characteristics, i.e., abilities and capabilities of the engineering resources in relation to knowledge.

# 2.4 Research gaps

In conclusion, the recent studies and, in particular, the most recent one by Cannas et al. (2019), can be considered very helpful to increase our understanding of engineering and production decoupling configurations in the ETO sector, as well as their performance outcomes. Cannas et al. (2019) considered all the previous literature, developing and empirically refining a 2D-CODP framework, which classifies the different order fulfilment strategies employed by ETO companies. Also, it categorised the performance outcomes of different 2D-CODP positioning choices. However, no attention was given in this study to the contextual factors influencing the choices of 2D-CODP positioning. In particular, this study, as well as the previous ones, did not consider the environment where the companies operate and the characteristics of their products and processes.

For this reason, our research is focused on this specific gap, i.e., the determinants for order fulfilment strategy choices. Regarding this issue, Table 1 summarises the different approaches found in the extant literature. The results show that, while recent work is very helpful in increasing our understanding of order fulfilment strategies in the ETO sector, there is still a gap between theory and practice when analysing engineering and production coordination to support ETO companies in adapting their order fulfilment strategy to the market (Mello et al., 2017; Sandrin et al., 2018). As transpires from the review of the literature and the synthesis in Table 1, in the CODP positioning field, there are still some limitations to be solved. The main focus of the recent studies has been on the customisation choices in the engineering process, with little attention to the interactions with production. As a consequence, the determinants for the positioning were mainly related to the product design and the market characteristics,

without deeply analysing the process dimension. This implies the need to consider the process characteristics, especially in terms of coordination, to understand which strategies are suitable to specific situations and avoid late design changes and production reworks (Mello et al., 2017, 2015a, 2015b).

None of the existing studies addresses a comprehensive understanding of the contingencies driving the 2D-CODP positioning, as previously done in the production-focused CODP literature. As a consequence, there is no guidance for ETO companies to understand what order fulfilment strategy to implement to suit different contextual factors. Therefore, there is a need for a profound analysis of the market, product and process characteristics. These characteristics, at the moment, represent for the ETO industry a "black-box" that requires to be opened in order to deeply understand the dynamics of 2D-CODP positioning. Consequently, our research aims at filling these gaps, by integrating the literature that explores the ETO strategies with the one related to the CODP concept, as suggested by Gosling and Naim (2009) and Dekkers et al. (2013). The goal is to build an initial framework of contingencies driving these applications and empirically explore it through multiple case studies research.

----- PLEASE INSERT TABLE 1 HERE -----

# 3 Methodology

This study applies an abductive approach that combines both theoretical and empirical insights. The goal of the study is to refine existing theory with new concepts, by interpreting an individual phenomenon in a contextual framework offering a new perspective (Kovács and Spens, 2005). Accordingly, the study follows a non-linear process: the results are obtained through the combination of literature and practice, through back and forth exchanges (Dubois and Gadde, 2002). Therefore, the literature review performed has been combined with direct empirical observations through case study research, which is considered a good method to refine theory based on empirical evidence (Eisenhardt, 1989). According to what the recent literature on case study research suggests (Hancock and Algozzine, 2016; Voss et al., 2016; Yin, 2018), the study has been conducted following 5 steps: first, we defined what we wanted to study based on the literature review, i.e., the research framework and constructs (subsection 3.1); second, we defined how we wanted to study it, i.e., the case study design (subsection 3.2); third, we defined whom we wanted to study, i.e., the case selection (subsection 3.3); fourth, we defined how to acquire the information, i.e., the data collection (subsection 3.4); fifth, we

defined how to analyse the information we acquired, i.e. the data analysis (subsection 3.5). Figure 1 summarises the steps followed in this research.

----- PLEASE INSERT FIGURE 1 HERE -----

# 3.1 Research framework and constructs

The first step in the case study research consisted of defining a "road map" of the route that the study aimed to follow, to be used as an aid for the researchers in collecting answers to the research questions (Hancock and Algozzine, 2016). This has been possible by means of the development of an organized research framework, presented in Figure 2 and Table 2. The research framework built a conceptual foundation for the empirical study and supported it in defining the concepts that were to be studied, i.e., the key constructs and variables, and the assumed relationships among them (Voss et al., 2016).

The literature review confirmed what the contingency theory claims: i.e., there is a need for a strategic fit with different types of environment (Lawrence and Lorsch, 1967; Luthans and Stewart, 1977; Thompson, 1967). The contingency perspective of the CODP positioning is based on the concept of "fit as moderation", which needs to be well clarified so that can be operationalised within the research framework and validated through empirical data (Venkatraman, 1989). According to the moderation perspective, the impact of the order fulfilment strategy implemented by the company (i.e., the predictor variable) on the performance outcome, (i.e., the criterion variable), depends on the alignment with a set of contextual variables (i.e., the moderator variable). Consequently, the constructs of the research framework, shown in Figure 2, include the main object of interest of the study, organised according to the classification of contingency theory models (Sousa and Voss, 2008).

Then, Table 2 lists these constructs and the variables to measure them, providing detailed definitions, and specifies the references based on the literature review presented in section 2. Reading them through a contingency theory lens, the order fulfilment strategy represents the strategic choice made by the company related to the 2D-CODP positioning: a set of strategies can be identified depending on the number of engineering and production activities performed after the order. As far as contextual factors concern, the market-related factors can be considered environmental variables that are not subject to direct control of the company and affect the system with which the company interacts. Whereas, product- and process-related factors can be defined as resource variables over which the company has direct control but that can change only in the long-term and with substantial efforts. Finally, the performance outcome represents the dependent measure that can be used to assess the fit between the contextual

factors and order fulfilment strategy and includes the traditional performance measured in the CODP literature: price, time, flexibility, reliability and innovativeness.

------ PLEASE INSERT FIGURE 2 HERE ------------ PLEASE INSERT TABLE 2 HERE ------

# 3.2 Case study design

The methodology applied in this study is a multiple case studies research. The exploratory aim of this study makes the collection of data from multiple cases preferable since it increases the external validity of the findings (Yin, 2018). However, the population addressed belongs to one industry in one country, i.e., the Italian machinery industry, to increase the control of variations within the population by limiting the research domain (Voss et al., 2016). The decision to address the Italian machinery industry was made for three main reasons. First, the companies operating in the machinery industry design and manufacture capital goods, which are considered as well representative of ETO issues (Adrodegari et al., 2015; Cannas et al., 2019; Dekkers, 2006; Veldman and Alblas, 2012). Second, Italian companies are among the world's top producers and exporters in the machinery industry, thanks to high customisation and flexibility, and a high level of technological innovation (source: Federmacchine, 2017). Third, the challenges that the Italian machinery industry faced in recent decades, due to globalisation, make this population representative of the research problem. In particular, the emerging markets of the Asia-Pacific region have had a big impact on the global competitive environment, offering low costs and short lead times with low added value. In 2016 (source: Goh, 2017), the percentage distribution of the global machinery production was about \$680 billion in the Asia-Pacific area (49.1%), \$406 billion in the European area (29.3%), and \$299 billion in the Americas (21.6%). The key players changed, and many companies in Italy, but also in the entire European area, according to McKinsey & Company (2016), are struggling in facing this global pressure and are starting asking themselves if the actual strategies applied in product development and production are sufficient to keep the pace of the recent developments.

#### 3.3 Case Selection

The complete list of companies operating in the Italian machinery industry was found in the database "AIDA" (<a href="https://aida.bvdinfo.com/">https://aida.bvdinfo.com/</a>) selecting only medium and big companies (according to EU recommendation 2003/361), to be sure that companies perform both engineering and production processes. The unit of analysis of the study is the product family

since a company can implement more than one strategy to provide the different product families to market.

Within this pre-defined list, case selection carefully followed theoretical reasons, based on a replication logic (Miles and Huberman, 1994): literal replication, i.e., companies were selected based on the expectation that they have similar contextual variables and performance outcomes since they offer product families that employ the same engineering and production decoupling configuration (i.e. the order fulfilment strategy); and theoretical replication, i.e., companies were selected based on the expectation that they have different contextual variables and performance outcomes, but for predictable reasons, i.e., they offer product families that employed a different order fulfilment strategy. Accordingly, in line with the main hypothesis of the study, the sample was selected for ensuring the maximum variation within the population and finding subgroups to compare and identify common patterns. This information was found in the reports developed by national industrial associations (Federmacchine, UCIMU, Amaplast, etc.), the Italian National Institute of Statistics (ISTAT), universities or consultancy reports as well as companies' websites. Moreover, the authors had experience in the machinery sector thanks to previous research projects conducted.

In total, eleven Italian machinery industry companies were selected, all recognised to be market leaders and located in Northern Italy, relevant geographical area in terms of impact on the total Italian machinery industry turnover. Eleven companies were considered sufficient, according to the replication logic applied, because they provided twenty-four cases and a good number of literal and theoretical replications, i.e., similar and rival order fulfilment strategies, which strength and support the findings of the study (Yin, 2018). In Table 3, the case study overview is presented. The percentage between brackets, next to the case studies, shows the total impact of the product analysed on the turnover.

----- PLEASE INSERT TABLE 3 HERE -----

#### 3.4 Data collection

The data collection was performed through multiple data collection methods, to ensure the presence of different sources of evidence (Stuart et al., 2002). The primary source was the face-to-face interview (source 1), always done by at least two researchers. The questionnaire has been organised as a semi-structured interview, to conduct the interview maintaining a logical order, while collecting open comments. The protocol was organised based on the constructs (i.e., order fulfilment strategy, contextual factors, performance outcome) defined in the research framework presented in subsection 3.1 (see Figure 2 and Table 2). Importance was

given to spontaneous deviations to preserve possible contradictory views with respect to the initial framework (Stake, 1995). The interviews were always recorded and transcribed. Based on the cultural context of the study, and the fact that both, the participants and the researchers that conducted the cases, spoke the same non-English native language, the interviews were conducted entirely in Italian. Therefore, maximum attention was given while translating from Italian to English to keep the validity of the findings (the details of the translation procedure are reported in the next section).

The participants were experts in the engineering and production processes (e.g. plant manager, operations manager, engineering manager, project manager), who knew the current situation and issues very well, and each case lasted on average 4 hours. Multiple interviews are missing in some cases because of constraints beyond the researchers' control, i.e., the availability of the company and its human resources. To overcome this issue and avoid the subjectivity typical of the single respondent (Voss et al., 2016) the use of different sources were essential: (i) direct observations (source 2) during plant and engineering department tours; (ii) company's website and official documents available in the web (source 3); (iii) product catalogue and other internal documents (source 4). The collection of this additional information helped in obtaining rigorous results thanks to the triangulation of evidence (Hays, 2004). The case study protocol used to collect data is shown in Appendix 1.

#### 3.5 Data analysis

Data were analysed by implementing a content analysis approach. The coding was executed manually by the authors: they independently read and coded the transcriptions, and then discussed the results to reach a common understanding of them. Finally, triangulation with additional insights from secondary data was performed in order to enrich the findings and overcome possible missing information. In this phase, maximum attention was given in keeping the meaning interpreted and communicated in the findings (English language) as close as possible to the meaning expressed by the interviewees (Italian language). According to the recommendations provided by van Nes et al. (2010), the data analysis was conducted in the original language as much as possible (first step coding), to avoid limitations in the analysis and reduce influences provided by studying the contents in another language. Secondly, when the translation was needed (second step coding and discussion of results), multiple in-depth discussions among the authors were performed to find the best translation, through fluid descriptions and various English formulations. It is important to underline that the research team included, since the beginning, one native English researcher, expert in the field.

The content analysis allowed us to explore the differences between the cases and search for patterns to compare with the initial tentative conceptual framework developed from the literature (Almutairi et al., 2014). The interviews and the documents were systematically analysed by evaluating the material and classifying the contents in the analytic categories taken from the literature and refined, enriched and extended, with additional insights coming from the empirical findings. In this sense, the main categories used to classify the literature were the ones related to the main constructs of the research framework (i.e., order fulfilment strategy, contextual factors, performance outcomes). A detailed description of all the cases within each construct and variable has been developed in structured tables.

The scoring method used for the analysis has been based on a relative and qualitative evaluation, with respect to the market analysed, as usual in qualitative research based on a narrow number of companies analysed. Given the exploratory nature, the final patterns could not be defined before the study but emerged from the collected data (Sinkovics, 2017). Therefore, the construct measures have been continuously validated during the study checking their consistency in the subsequent interviews and calling back the companies, if any unreliable measure emerged. In this way, the researchers assured the coherence of the analysis and the possibility to reach general conclusions.

Finally, to ensure the maximum transparency, the paper illustrates the chains of evidence from raw data to interpretation through a careful documentation of the data analysis, provided in Appendix 2, which shows a complete example of the data analysis performed for case H1, and in the next section 4, which deeply reports the results from the cases, with the support of interviews quotes (Caniato et al., 2018; Yin, 2018).

# 4 Order fulfilment strategy and contextual factors alignment

# 4.1 Understanding order fulfilment strategies across the ETO case studies

The first output of the data analysis is the classification of the case studies according to the order fulfilment strategy implemented. Within our case studies, the data suggest that the companies implemented different order fulfilment strategies to provide their product families to the market. Figure 3 shows the final classification of the cases according to the number of engineering and production activities performed to order. The classification is based on the 2D-CODP positioning matrix proposed by Cannas et al. (2019) for the classification of product families provided by companies operating in the machinery industry. This choice was made because Cannas et al. (2019) framework, as anticipated in the literature review section, can be

considered the most recent and comprehensive to date. Indeed, it classifies all the previous literature on ETO order fulfilment strategies (as, for example, Dekkers, 2006; Gosling et al., 2017; Wikner and Rudberg, 2005; Willner et al., 2016) and includes new decoupling strategies identified through an extended qualitative field study in the machinery industry.

The engineering process activities to perform can be divided into two main sub-processes: (i) research and development, which involves the research, test and prototype of a new product concept; (ii) design of product Bill of Material (BoM), including the list of components required to build the machine along with the technical specifications, the instructions for production processes and the instructions for final assembly. In the cases analysed, research and development are always performed to forecast, as well as the design of the "general" BoM, because the new products are first presented to industrial fairs (Plast, BI-MU, etc.). Adaptations of the product BoM can then be performed completely to order (with major modifications, adapting the existing components to order, or eliminating/adding components), partially to forecast (with minor modifications or only final combinations to order) or completely to forecast (by taking the design as it is). Whereas, the production process activities to perform can be divided into three main sub-processes: (i) purchase, which involves the procurement of the raw materials; (ii) manufacturing, which involves the physical realisation of the components, either produced internally or outsourced to specialised suppliers; and (iii) assembly, which involves the assembly of the components. In the cases studied, the procurement of the raw materials is always performed to forecast. The manufacturing and assembly, instead, can be performed completely to order (make to order), partially to forecast (finalise the production/procurement of customised components to order or assemble to order), or completely to forecast (make to stock).

Five different order fulfilment strategies have been identified. They range from "special" to "standard". It can be seen in Figure 3 that standard machines represent an unusual strategy in this sector, only two cases were related to this strategy, related to a small percentage of the product portfolio. Also, special machines are related to a small number of cases, only three, but two out of three are related to the more than 90% of the product portfolio. While intermediate strategies, i.e., "customised", "standard-customised" and "modular", are the most applied ones. Product family F4 is characterised by "special" machines, vice versa, product family F3, is characterised by "standard" machines. According to the plant manager of company F, "F4 is the most customised case. The engineering department, in collaboration with the sales, develops the concept of the machine starting from the catalogue but applying technical modification to more than 70% of the product BoM and producing all the components

to order [...] In the case of F3, instead, the design is taken as it is from the product catalogue and the production and assembly activities are planned to forecast". Product family H1, already described in Appendix 2, is characterised by "customised" machines. This implies that the engineering activities after the order consist in a major modification of some existing components (30% on average, in this specific case) while manufacturing activities can only partially start before the order, by purchasing or producing the strategic components. While the order fulfilment strategy for products belonging to product family such as H2 is "standardcustomised". Meaning that the products are already designed and introduced in the product portfolio but slightly modified after the order to achieve specific requirements. As a consequence, according to the plant manager of company H interviewed, "the minor modifications of the design allow the company to produce all the components of the BoM to forecast and producing or adjusting the customised components to order". Finally, product family H3 is characterised by "modular" products, meaning that they are already designed and introduced in the product portfolio and only finalised to order based on the combination of the design options to obtain the desired BoM configuration. As a consequence, according to the plant manager of company H interviewed, "thanks to the high engineering standardisation level, in this case, the company prefers to perform the manufacturing activities of the components completely to forecast".

----- PLEASE INSERT FIGURE 3 HERE -----

# 4.2 Market-related factors

The evidence of the cases shows that the order fulfilment strategies implemented by the companies are aligned to the different customer typologies. One of the factors characterising the different customer typologies is the knowledge, which has been measured by all the companies as technical knowledge, i.e., experience and know-how in the actual development of the machine, and functional knowledge, i.e., experience and know-how in the utilisation of the machine, in a qualitative rating scale from "high" to "low/none". "If the customer has low technical knowledge, we have a higher degree of freedom in developing the machine using the existing technical specification and we can reduce the number of engineering and production activities performed to order" (D's engineering manager). Another factor to consider when companies align the order fulfilment strategy to the market is the customer size. The customer size has been measured by all the companies distinguishing from multinational companies and medium/small subcontractors. "There are two types of customers, big customers, which are multinational companies that demand high flexibility and customisation but accept premium

prices and longer delivery lead times, and the smaller, cost-conscious subcontractors, which want a quick product delivery to satisfy a current market trend" (E's sales manager). Finally, the demand variety represents an important market factor to consider. The demand variety has been measured by all companies as a percentage, by evaluating how many customers — out of ten — on average, ask for changes after the order. "We have customers that often change their mind after the order. In this case, it is better to wait for the order before to perform the engineering and production activities, otherwise, we risk to rework the components many times" (G's engineering manager).

In particular, as shown in Table 4, to respond to knowledgeable and demanding multinational customers, it is necessary to implement special or customised strategies. In fact, they leverage the flexibility they still have on design and production to accommodate customer needs. Vice versa, standard and modular strategies are implemented for subcontractors that choose the product from the catalogue looking for the desired product functionalities, having little or none technical knowledge. In the case F4, for example, "The customer is a big customer, with high technical know-how and experience related to the product, which defined the desired technical specifications and often asks for changes. 8-9 customers out of 10 require for changes after the order. [...] In the case F3, customers are small subcontractors not specialised in specific production activity, because they depend on the market trends. They have little (or none) experience and knowledge related to the product and express only a preference for general functionalities. In this case, the probability for late changes is almost null: we provide F3 as it is, without any changes" (F's plant manager). There are also intermediate cases, such as H2, where "the customers interested in the product family are mixed, both big multinational companies and small subcontractors, which search for highprecision machines in short lead time and at a competitive price. They ask for some technical specifications but mainly expressing preferences for specific functionalities. In this case, on average, 4-5 customers out of 10 require changes after the order" (H's plant manager).

----- PLEASE INSERT TABLE 4 HERE -----

# 4.3 Product-related factors

Table 5 shows the results obtained from the analysis of the cases in terms of product characteristics. When talking about the product, the technology life cycle was of significant interest to the cases analysed. All the interviewed people refer to it as: new/emerging, growing or mature. "New technologies make necessary to keep high customisation since they are still not as stable as the mature ones, and it is difficult to forecast the customer needs" (A's

engineering manager). Family F4, for example, is characterised by new/emerging technologies, "the customer asks for new technical characteristics and functionalities. The innovation typically consists of increasing the production capacity with the same machine. This means to find new geometries, to improve materials, to change components, etc. Thus, it is highly certain that new components and major modifications will be added, and it is better to wait for the customer order to produce the product" (F's plant manager). While family H3 is related to mature and stable technologies. "H3 demand is easy to forecast and the risk of uncertainty is low. This makes possible to rely on design modularity and combination of modules, in case of changes in the design requirements." (H's plant manager).

Moreover, the customisation opportunities impact to the order fulfilment strategy, and it has been measured by all the companies as the range of choices provided to the customer: only catalogue with standard products; catalogue and modular options (fundamental characteristics of the product that increase the perception of variety); catalogue, modular and additional options (not fully needed but make the customer satisfied); catalogue, modular, additional and superior options (delighters that exceed customer expectations); complete customisation with no catalogue or catalogue only as a guideline. "If the variety in terms of customisation opportunities is high, the company increases the range of choices for the customer. Thus, it is hard to forecast the demand and keep stock of components and spare parts, because the stock holding cost would be excessive. While the reduction of customisation opportunities decreases the variety but also the uncertainty" (D's engineering manager). For instance, the product family H1 is composed of 8 different products in the catalogue, H2 of 6 products and H3 of 2 products. Each of the products is proposed with different options. To confirm this, the plant manager showed a set of internal documents containing the technical specifications (i.e., the different design options proposed to the customer when the order arrives): "We propose to the customer different options: modular options, additional options, and superior options. As you can see, the H1 document is about 90 pages characterised by all modular, additional and superior options; H2 document is about 50 pages characterised by modular and additional options; and, finally, H3 document is 20 pages characterised by only modular options". In case of F3 and F4, "F3 is composed of only two different machines without any option/variants, provided as they are in the catalogue with their fundamental characteristics. While F4 has almost infinite customisation opportunities, the only constraint is the milling technology. In this case, we use the catalogue only as a guideline and we fulfil the order exploiting experience and knowledge" (F's plant manager).

Finally, the modular product structure represents an important factor. The companies refer to it as non-existent, partial or complete. The modular product structure was identified as a common characteristic of all the product families offered by company H. In particular, H1 and H2 are partially modular and H3 completely modular. All the companies that have intermediate configurations (i.e., customised, standard-customised and modular machines) leverage on modules to anticipate some engineering and production activity before the order and adapt the remaining ones to order. Interestingly, all cases of the pure customised (i.e., special machines) and pure standardised (i.e., standard machines) configurations have been identified as nonmodular. In these cases, modularity is not considered worthy because, as stated by company F, "the level of design modularity helps to increase production responsiveness, but it also increases cost. This is because it is necessary to (at least) duplicate the production activities but use only a part of what you produced. And, if the customer order is always different, I could produce 100 pieces with 48 holes each, to have the highest possible modular interface in the assembly activities, but then, I exploit only 2 holes for each order. In this case, I decided to reduce the modular configurability, producing the holes to order, with longer lead times but saving a lot of money".

----- PLEASE INSERT TABLE 5 HERE -----

# 4.4 Process-related factors

In Table 6, the list of process factors related to each different order fulfilment strategy is provided. The cases analysed defined as important factors affecting the order fulfilment strategy choice, the sales and engineering processes structure, the production flexibility and the sales, engineering and production interface.

All case studies highlight that the sales and engineering processes structure is related to the degree of freedom given to the resources in proposing solutions. A qualitative scale has been defined, ranging from "loosely structured process with high degrees of freedom" to "process guided by a procedure with no degrees of freedom". In this sense, data show that the higher the customisation the looser the structure of the sales and engineering processes. "The engineering process of product F4 does not follow any specific procedures so that it is possible to completely meet the customer requirements. Therefore, the salesmen and the engineers act with a high degree of freedom but being aware of the impact of their decisions on the other supply chain processes and the entire value chain. Vice versa, in case of F3, sales and engineering departments deal with standard machines, based on mature technology. Therefore, they apply rigorous procedures, with no need to consider the impact of their choices on the other

departments" (F's plant manager). In the intermediate strategies, instead, there is always a procedure, but with different degrees of freedom. In this sense, the plant manager of company H showed to us the internal procedure followed by the sales and engineering departments: "As you can see it is divided into various sections: the performance that the machine must guarantee, the technical characteristics, the sales plan, the list of the various configurations already provided to the market. These different configurations are already planned and proposed to the customer and each of them corresponds to a specific budget of costs and lead times. This document helps the salesmen and engineers to understand the impact of each machine configuration to the entire supply chain. Then, in the case of H1, the choice is to keep a high degree of freedom in changing the initial characteristics, being aware of the impact on the value chain. This freedom is lower for H2 and null for H3".

All the companies consider production flexibility as the capability to manage unexpected requirements for reworks in a short time after the order. They use a qualitative scale to assess this factor, comparing them with the market average and following this classification: "very quick in managing reworks", "quick in managing reworks", "managing reworks with negative impact on delivery lead times", or "not managing reworks". This factor resulted to be very important in case of high customised strategies. "We are accustomed to late and unexpected design changes and production reworks. Therefore, we organise our production activities so to be flexible and ready to face reworks very quickly" (I's production manager). Instead, the standard and modular order fulfilment strategies are less ready to face reworks and the impact on lead times is higher. For example, in the case H3 "the company completes all the procurement and production activities of the modules before the customer order. This reduces the production flexibility and means that in case of special requirements included in the customer order what has been purchased/produced is not suitable anymore. The reworks heavily affect the delivery lead time".

Finally, sales, engineering and production interface is a relevant factor that the companies measured as the need for exchanges of information and warning and synchronous communication. They refer to it by using a qualitative rating that ranges from "continuous need" to "limited" to "sporadic". An important finding is that the higher is the customisation, the more important is to continuously engage the interfaces and synergy among the departments and to employ synchronous communication (i.e., simultaneously and in real time, such as face to face meetings or calls) after the order. For example, according to case F "improving the interface between the sales, engineering and production departments is

fundamental in the case F4, to facilitate activities overlapping and reduce delivery lead times. For example, we leverage on a real-time data sharing between the functions, important to increase the visibility and traceability of the information related to the order management. This implies a reduction of errors and, consequently, a reduction of costs and lead times thanks to real time problem solving, workloads balancing optimisation, etc." (F's plant manager). Whereas, in case of high standardisation the interfaces among the departments after the order are sporadic and asynchronous (i.e., not live and deferred, such as emails).

----- PLEASE INSERT TABLE 6 HERE -----

# 5 Performance outcomes and strategic fit

The insights from the case study research confirm the initial assumption that ETO companies focus their efforts in carefully aligning their order fulfilment strategy to the contextual factors to achieve the desired performance outcomes. The performance identified as relevant outcome of the strategic alignment between strategy and context are time, price, flexibility, reliability and innovativeness, and the companies interviewed assess them, referring to each product family, with respect to the market average (0 – not competitive, 1 – low competitive, 2 – on market average, 3 – competitive, 4 – very competitive).

If the fully customised strategy (i.e., special machines) is implemented to address the right market segment (i.e., niche market, composed by knowledgeable and demanding customers), supported by the suitable product (i.e., innovative and customised product) and processes (i.e., loosely structured but highly flexible and strongly aligned), the final performance outcome expected is the capability to always meet the customer requirements for customisation, i.e., high flexibility, and provide them with high innovativeness in terms of technologies. This justifies a premium price and longer delivery lead times, compared with the competitors. Moreover, the customer accepts the risk of unexpected defects, because the number of errors increases in new designs and there is a lack of previous feedbacks from the production department and suppliers. "When the customer asks for F4, asks for a new machine model, a special machine. The competition is on technology and flexibility. This is the most complicated case; the machine does not exist yet. And, of course, already in the initial phase the sales department leverages the engineering department to understand the machine, the feasibility, etc. Therefore, the costs will be higher, and all the subsequent steps will be much slower, making the delivery lead time longer than 10 months. For example, the purchasing department may have to buy things for the first time, there are new production cycles, etc."

Vice versa, the standard machines are addressed to a stable market (i.e., predictable customers with no specific requests) that must be served with mature and standardised products and supported by not flexible but highly structured processes. This alignment aims to assure the capability to achieve high performance in terms of price, delivery lead times and reliability, at the expenses of flexibility, because customer requirements for customisation are never met, and innovativeness, because the technology is already mature. "Product family F3 addresses a market segment where the product is not required with special technological skills. Instead, it opens up a world of competitors larger and more aggressive than F4. The competition is no longer on technology but on price and delivery lead time, as well as reliability. We exploit our standard product configurations and standard procedures to achieve this goal." (F's plant manager).

Finally, intermediate strategies allow companies to achieve a good compromise between different competitive proprieties. For example, H1 has a dominant market positioning, namely, as the plant manager explained, "it is a leader in a specialised market, where it is distinguished from competitors thanks to the possibility to provide state-of-the-art technologies that satisfy almost all customer requirements". The price and delivery lead time are higher than competitors but reduced at minimum, only to justify the additional customised components: price 20% higher than the market average and delivery lead time 5-8 months, depending on the extent and complexity of design modifications required by the customer. The risk of unexpected defects is related only to the components strongly modified, usually not more than 30% of the product BoM. H2 has a strong market positioning, as the plant manager specified: "H2 is among the leaders in a specialised market, thanks to the capability to provide customers with economic complete machines with state-of-the-art technologies". H2 partially meets the customer requirements for customisation and the price is competitive and comparable to the market average. Additionally, the delivery lead time is 2-4 months, based on the extent of design modifications required by the customer and the risk for defects is contained since the changes applied to the existing design are only minor. Finally, H3, as the plant manager specified: "is a follower in a specialised market, looking at the first three leading companies. It tries to accomplish the same product quality and precision, at competitive prices, for customers that need conventional machines in short lead times, 1-2 weeks at maximum". H3 rarely meets requirements for customisation and the innovativeness is little, but still offers variety concerning general personalisation, thanks to modularity. Also, the risk for defects is almost null because it exploits mature technologies.

Figure 4 shows in detail the main results observed in the cases to help readers better understand differences performance outcomes of the range of order fulfilment strategies within the same ETO industry, when well aligned to the context.

----- PLEASE INSERT FIGURE 4 HERE -----

#### 6 Discussion

This study empirically investigated the order fulfilment strategies implemented in eleven ETO companies through a contingency theory lens. Twenty-four different product families and five different order fulfilment strategies have been identified, i.e., standard machines, modular machines, standard customised machines, customised machines, special machines. For each order fulfilment strategy, the sources of differentiation between the environments have been defined and structured within the market, product and processes constructs, answering to RQ1. Then, the performance outcome of each strategy has been measured, explaining the rationale for the positioning of the product families in different 2D-CODP locations, answering to RQ2.

6.1 Answering RQ1: What are the determinants for the order fulfilment strategy choices within ETO companies?

The results shown that the determinants for order fulfilment strategy choices within ETO companies are contextual factors characterising the market (i.e., customer product knowledge, customer size, and demand variety), the product (i.e., technology life-cycle, customisation opportunities, and modular design), and the processes (i.e., sales and engineering process structure, production flexibility, and sales, engineering and production interface).

The results confirm the validity and meaningfulness of classifying the determinants into market and product characteristics, as suggested by past CODP studies (Hallgren and Olhager, 2006; Olhager, 2003), and process characteristics as recently underlined by 2D-CODP and ETO studies (Dekkers, 2006; Gosling et al., 2017; Schoenwitz et al., 2017; Wikner and Rudberg, 2005). By opening the "black-box" of the market, product and process related factors, the existing literature has been refined to a greater level of detail for the ETO sector. Indeed, many factors identified and classified in Table 1 according to the three different literature streams (production, supply chain management, ETO), were initially considered too general to become meaningful in distinguishing the different strategies employed by the companies analysed. Therefore, the study built on the existing literature but extended it, by revisiting the existing determinants and developing additional factors, as explained below, to cater for the ETO context.

As far as market-related factors concern, the average annual units sold and the market growth or changes, as introduced by the recent ETO literature, resulted too generic to describe and classify the market contextual factors of the companies analysed. Whereas, the characteristics typically related to ETO and agile configurations by the production and supply chain management literature, i.e., unpredictable demand, low volumes, wide product range, and high customisation requirements, were confirmed as existent by all the companies analysed, for all the product families. Thus, they did not result as determinants for order fulfilment strategies choices in the machinery industry; rather, they appear as qualifiers of the ETO sector where they operate. The companies interviewed underlined that their customers expect to be given high room for customisation and a wide range of choices. However, customers are diverse and have very different ways to interact with the machinery manufacturer when placing an order. In particular, they can be profiled in different groups based on three characteristics: (i) their knowledge regarding the product, (ii) their size, and (iii) their requirements for changes after the order. For this reason, the definition of a portrait of customers based on these characteristics is considered fundamental to make the strategic decision of the order fulfilment strategy for a product family. Therefore, as result of this research, new factors were introduced, mostly related to the characteristics of every single customer: (i) the customer technical and functional product knowledge; (ii) the customer size; and (iii), the customer requirements for late changes after the order (demand variety).

As far as product-related factors concern, all the product families analysed in the study had complex deep structures. Thus, these two characteristics proposed by the production literature were not considered determinants for order fulfilment strategies choices by the companies analysed; rather, they are considered peculiarities of the ETO products provided in the machinery industry. However, the other determinants did not differ from the literature, building on it and confirming the contingency of three variables: (i) modularity, as suggested by production and ETO literature; (ii) technology life-cycle (new technologies – innovative products vs mature technologies – functional products), as suggested by the supply chain management literature; and (iii), the customisation opportunities offered to the customers, as suggested by the ETO literature.

As far as process-related factors concern, on the one hand, the results differed from the literature regarding variables such as the number of planning points and bottlenecks, as well as the abilities and capabilities of the engineering resources, which were not considered by the companies analysed as characteristics that determine their order fulfilment strategy choices. On the other hand, the findings confirmed the literature and built on the determinant

"production flexibility". If the production department is flexible to solve very quickly requirements for reworks, high-customised order fulfilment strategies can be employed; whereas if it takes an increasing amount of time to react to unexpected changes, the standardisation of the order fulfilment strategy should increase. Also, new interesting insights emerged. First, the presence of standard procedures for the upstream processes (sales and engineering) was underlined as an important determinant for order fulfilment strategy choices. Indeed, if the degree of freedom is high when proposing and designing the solution, i.e., the sales and engineering processes are loosely structured, it is an ideal condition for special machines, but it makes very difficult to introduce standards in the products. Second, continuous inter-functional interfaces after the order (i.e., the exchange of information and warnings and synchronous communication) were underlined as not necessary when standard machines are provided, but very important in the case of special machines.

# 6.2 Answering RQ2: How can ETO companies fit the order fulfilment strategies to contextual factors?

The findings from the cases demonstrated that the values of the determinants for order fulfilment strategies gradually change based on the customisation level of the order fulfilment strategy employed. From high standardised to high customised fulfilment strategies: (i) the market-related factors range from predictable customers with no specific requests to knowledgeable and demanding customers; (ii) the product-related factors range from mature and standardised products to innovative and customised products; (iii) the process-related factors range from highly structured, loosely flexible and not aligned processes and loosely structured, highly flexible and strongly aligned process. The fit of these characteristics with the suitable order fulfilment strategy has been demonstrated to provide the desired performance outcomes, which range from very competitive price, time and reliability for high standardised strategies to very competitive flexibility and innovativeness for high customised ones.

By comparing the empirical results with the literature, we confirm the existence of a tradeoff between performance outcomes shifting from high standardised to high customised strategies, as underlined over years by several production and ETO studies (Barlow et al., 2003; Cannas et al., 2019; Chase et al., 2001; Gosling et al., 2017; Rudberg and Wikner, 2004). Also, the cases confirm the current paradox of ETO companies aiming at multiple conflicting competitive priorities by implementing hybrid strategies and achieve the benefits of mass customisation (Mello et al., 2017; Sandrin et al., 2018). Additionally, the study brought innovativeness to the previous literature, by adding the reasons for forward and backwards shifting of the 2D-CODP. As already underlined in the literature review, this topic was underresearched by the previous literature, which deeply explored the dynamics of the shifting only in the one-dimensional framework (Hallgren and Olhager, 2006; Olhager, 2003).

The results show that the companies analysed fit existing environments to strengthen specific competitive priorities and achieve definite market positioning (i.e., niche, dominant, strong, follower or stable). In particular, the 2D-CODP positioning requires to carefully consider the interfaces between engineering and production. In fact, the production standardisation increases hand in hand with engineering: the case companies leverage engineering standardisation to exploit economies of scale and learning curves in the sales, procurement and production activities, implementing production strategies different from MTO with competitive costs and lead times. The efforts made, in this sense, is mainly related to: (i) defining a product catalogue supported by modular design architectures and technology development efforts; (ii) investing in procedures to guide both the sales and engineering processes, without losing too much production flexibility and coordination between the functions; (iii) facing new barriers in the market due to changes in customer characteristics, and the possible entrance of new customers.

#### 7 Conclusion

The purpose of this paper was to better understand the determinants of order fulfilment strategies for ETO companies, and, via our empirical study, to give insights into how organisations in the machinery sector react to strategically fit the order fulfilment strategy to their environment in order to achieve the desired performance outcomes. The overview of the performance outcome, provided in Figure 4, is one of the main outputs of this paper, which shows the determinants (i.e., the contextual factors) for order fulfilment strategies in one specific ETO industry (i.e., the machinery industry). It also indicates the effects that the strategic fit of the order fulfilment strategies with the contextual factors characterising the market, product and processes of the companies analysed has on performance.

The contributions of this paper are both theoretical and empirical. From a theoretical point of view, this paper increases the understanding of the determinants for order fulfilment strategies in ETO companies, and the reasoning for shifting from one strategy to another, exploring contingencies affecting the CODP positioning in the two-dimensional framework recently introduced by Cannas et al. (2019). Very little research has addressed these issues in the past, even if recent studies underlined the importance to explore the topic, deepen and refine

the outdated production literature, by addressing empirical data in the ETO area (Gosling et al., 2017; Willner et al., 2016). Previous studies in the ETO literature supported the idea of the "one-size-does-not-fit-all" and the importance of the strategic fit in stable and closed MTS context, but none of them provided a rich description of contingencies affecting this choice in the ETO context. The discussion of the findings in this paper shows how the cases provide new insights, building on the literature and introducing new interesting concepts, that progress the theory and knowledge of the ETO realities. The market-, product- and production-related factors identified in the literature have been enriched with the inclusion of the engineering perspective and its integration with production. Also, the company perspective has been substituted with the product family perspective, including the possibility of multiple strategies in the same company. This increases the effectiveness of the analysis in representing reality and reducing the gap between theory and practice. In conclusion, these results further the ETO research, renovating the seminal studies on CODP positioning (Hallgren and Olhager, 2006; Olhager, 2003) and providing foundations for future research on this topic.

From an empirical point of view, this study presents an example of the application of the 2D-CODP framework to the machinery industry, providing empirical evidence of its power in representing the order fulfilment strategies employed in real ETO contexts. Also, the paper shows practical examples of how companies leading different market segments in the machinery industry define their order fulfilment strategies. The results show that this strategic choice is not pre-defined and unique, and the best optimal strategy does not exist. This can be translated into a practical tool for companies. In fact, the cases analysed exemplifies how the contingency-based framework can be practically used, opening the "black box" of the theoretical constructs and providing practical examples of what are the contextual factors suitable to each strategy and the performance outcomes expectations. Moreover, the empirical study shows how the different order fulfilment strategies can be compared, providing a model for strategic decision-making. This paper is a first attempt to help ETO managers in understanding if they are implementing the suitable strategy for achieving the desired performance outcome and what are the key resources that must be improved when facing changes in their competitive environment, to align the organizational goals to new desirable performance. This contribution supports decision making in the challenging phenomenon of ETO companies that are pursuing the achievement of good performance in multiple conflicting competitive priorities shifting towards mass customisation strategies.

Finally, although this study has been carefully designed taking into consideration all the essential aspects for assuring high qualitative outcomes, some limitations exist, which are not

under the control of the researchers, opening interesting opportunities for further research. The case study research is a qualitative approach, primarily exploratory. On the one hand, this method helped the researchers for building and refining the extant theory; on the other hand, it limited the possibility to use measurable data to quantify the problem and perform statistical analysis. For this reason, the application of quantitative methodologies to the framework developed in this study is considered a relevant opportunity for future research, helping to validate the findings obtained. Also, even if the conceptual framework has been developed considering literature studies conducted in other industries, this study assessed it only in one specific industry and one specific country. As a consequence, further research is welcome to endorse and extend the findings of this research, by addressing other ETO sectors.

#### References

- Adrodegari, F., Bacchetti, A., Pinto, R., Pirola, F., Zanardini, M., 2015. Engineer-to-order (ETO) production planning and control: An empirical framework for machinery-building companies. Prod. Plan. Control 26, 910–932. https://doi.org/10.1080/09537287.2014.1001808
- Aitken, J., Christopher, M., Towill, D., 2002. Understanding, Implementing and Exploiting Agility and Leanness. Int. J. Logist. Res. Appl. 5, 59–74. https://doi.org/10.1080/13675560110084139
- Almutairi, A.F., Gardner, G.E., McCarthy, A., 2014. Practical guidance for the use of a pattern-matching technique in case-study research: A case presentation. Nurs. Health Sci. 16, 239–244.
- Amaro, G., Hendry, L., Kingsman, B., 1999. Competitive advantage, customisation and a new taxonomy for non make-to-stock companies. Int. J. Oper. Prod. Manag. https://doi.org/10.1108/01443579910254213
- André, S., Elgh, F., 2018. Modeling of transdisciplinary engineering assets using the design platform approach for improved customization ability. Adv. Eng. Informatics 38, 277–290.
- Barlow, J., Childerhouse, P., Gann, D., Hong-Minh, S., Naim, M., Ozaki, R., 2003. Choice and delivery in housebuilding: Lessons from Japan for UK housebuilders. Build. Res. Inf. 31, 134–145. https://doi.org/10.1080/09613210302003
- Birkie, S.E., Trucco, P., 2016. Understanding dynamism and complexity factors in engineer-to-order and their influence on lean implementation strategy. Prod. Plan. Control 27, 345–

- 359. https://doi.org/10.1080/09537287.2015.1127446
- Bozarth, C., Chapman, S., 1996. A contingency view of time-based competition for manufacturers. Int. J. Oper. Prod. Manag. 16, 56–67. https://doi.org/10.1108/01443579610119090
- Caniato, F., Doran, D., Sousa, R., Boer, H., 2018. Designing and developing OM research—from concept to publication. Int. J. Oper. Prod. Manag. https://doi.org/10.1108/IJOPM-01-2017-0038 Permanent
- Cannas, V.G., Gosling, J., Pero, M., Rossi, T., 2019. Engineering and production decoupling configurations: An empirical study in the machinery industry. Int. J. Prod. Econ. 216, 173–189.
- Cannas, V.G., Pero, M., Rossi, T., Gosling, J., 2018. Integrate Customer Order Decoupling Point and Mass Customisation Concepts: A Literature Review, in: Customization 4.0. Springer, pp. 495–517. https://doi.org/10.1007/978-3-319-77556-2 31
- Caron, F., Fiore, A., 1995. "Engineer to order" companies: how to integrate manufacturing and innovative processes. Int. J. Proj. Manag. 13, 313–319. https://doi.org/10.1016/0263-7863(95)00023-J
- Chase, R.B., Aquilano, N.J., Jacobs, F.R., 2001. Operations management for competitive advantage, 9th ed. McGraw-Hill, Boston, MA.
- Christopher, M., Peck, H., Towill, D., 2006. A taxonomy for selecting global supply chain strategies. Int. J. Logist. Manag. 17, 277–287.
- Christopher, M., Towill, D., 2001. An integrated model for the design of agile supply chains. Int. J. Phys. Distrib. Logist. Manag. 31, 235–246.
- Christopher, M., Towill, D.R., 2002. Developing market specific supply chain strategies. Int. J. Logist. Manag. 13, 1–14.
- Christopher, M., Towill, D.R., 2000. Supply chain migration from lean and functional to agile and customised. Supply Chain Manag. An Int. J. 5, 206–213.
- Davis, S.M., 1989. From "future perfect": Mass customizing. Plan. Rev. 17, 16–21.
- Dekkers, R., 2006. Engineering management and the order entry point. Int. J. Prod. Res. 44, 4011–4025. https://doi.org/10.1080/00207540600696328
- Dekkers, R., Chang, C.M., Kreutzfeldt, J., 2013. The interface between product design and engineering and manufacturing: A review of the literature and empirical evidence. Int. J. Prod. Econ. 144, 316–333. https://doi.org/10.1016/j.ijpe.2013.02.020
- Dubois, A., Gadde, L.E., 2002. Systematic combining: An abductive approach to case research.

  J. Bus. Res. 55, 553–560. https://doi.org/10.1016/S0148-2963(00)00195-8

- Eisenhardt, K.M., 1989. Building Theories from Case Study Research. Acad. Manag. Rev. 14, 532–550. https://doi.org/10.5465/AMR.1989.4308385
- Fisher, M., 1997. What is the Right Supply Chain for Your Product? Harv. Bus. Rev. https://doi.org/Article
- Flyvbjerg, B., 2014. What you should know about megaprojects and why: An overview. Proj. Manag. J. 45, 6–19.
- Giesberts, P.M.J., van der Tang, L., 1992. Dynamics of the customer order decoupling point: Impact on information systems for production control. Prod. Plan. Control. https://doi.org/10.1080/09537289208919402
- Gosling, J., Hewlett, B., Naim, M.M., 2017. Extending customer order penetration concepts to engineering designs. Int. J. Oper. Prod. Manag. 37, 402–422. https://doi.org/10.1108/IJOPM-07-2015-0453
- Gosling, J., Naim, M.M., 2009. Engineer-to-order supply chain management: A literature review and research agenda. Int. J. Prod. Econ. 122, 741–754. https://doi.org/10.1016/j.ijpe.2009.07.002
- Hallgren, M., Olhager, J., 2006. Differentiating manufacturing focus. Int. J. Prod. Res. 44, 3863–3878. https://doi.org/10.1080/00207540600702290
- Hancock, D.R., Algozzine, B., 2016. Doing case study research: A practical guide for beginning researchers. Teachers College Press.
- Haug, A., Ladeby, K., Edwards, K., 2009. From engineer-to-order to mass customization. Manag. Res. News. https://doi.org/10.1108/01409170910965233
- Hays, P.A., 2004. Case study research, in: deMarrais, K., Lapan, S.D. (Eds.), Foundations for Research: Methods of Inquiry in Education and the Social Sciences. Lawrence Erlbaum Associates, Inc., pp. 217–234.
- Hedenstierna, P., Ng, A.H.C., 2011. Dynamic implications of customer order decoupling point positioning. J. Manuf. Technol. Manag. 22, 1032–1042.
- Hicks, C., Mcgovern, T., Earl, C.F., 2000. Supply Chain Management A Strategic Issue In Engineer To Order Manufacturing. Int. J. Prod. Econ. 65, 179–190. https://doi.org/10.1016/S0925-5273(99)00026-2
- Hicks, C., McGovern, T., Earl, C.F., 2001. A typology of UK engineer-to-order companies. Int. J. Logist. 4, 43–56.
- Hill, T., 1993. Manufacturing strategy: the strategic management of the manufacturing function, Open University set book. Macmillan.
- Hoekstra, S., Romme, J., 1992. Integral Logistic Structures: Developing Customer-oriented

- Goods Flow, Industrial Press.
- Johansson, J., Elgh, F., 2019. Knowledge Objects Enable Mass-Individualization, in: Evolutionary and Deterministic Methods for Design Optimization and Control With Applications to Industrial and Societal Problems. Springer, pp. 371–386.
- Johnsen, S.M., Hvam, L., 2018. Understanding the impact of non-standard customisations in an engineer-to-order context: A case study. Int. J. Prod. Res. 1–15.
- Jonsson, H., Rudberg, M., 2014. Classification of production systems for industrialized building: A production strategy perspective. Constr. Manag. Econ. 32, 53–69. https://doi.org/10.1080/01446193.2013.812226
- Konijnendijk, P.A., 1993. Dependence and conflict between production and sales. Ind. Mark. Manag. 22, 161–167. https://doi.org/10.1016/0019-8501(93)90001-N
- Kovács, G., Spens, K.M., 2005. Abductive reasoning in logistics research. Int. J. Phys. Distrib. Logist. Manag. 35, 132–144.
- Kristianto, Y., Helo, P., Jiao, R.J., 2013. Mass customization design of engineer-to-order products using Benders' decomposition and bi-level stochastic programming. J. Intell. Manuf. 24, 961–975.
- Kristjansdottir, K., Sha, S., Hvam, L., Bonev, M., Myrodia, A., 2018. Return on investment from the use of product con fi guration systems A case study. Comput. Ind. 100, 57–69. https://doi.org/10.1016/j.compind.2018.04.003
- Kritchanchai, D., MacCarthy, B.L., 1999. Responsiveness of the order fulfilment process. Int. J. Oper. Prod. Manag. 19, 812–833. https://doi.org/10.1108/01443579910274419
- Lambert, D.M., Cooper, M.C., 2000. Issues in supply chain management. Ind. Mark. Manag. 29, 65–83.
- Lampel, J., Mintzberg, H., 1996. Customizing Customization. Sloan Manage. Rev. 38, 21–30. https://doi.org/10.1002/dir.20076
- Lawrence, P.R., Lorsch, J.W., 1967. Differentiation and integration in complex organizations. Adm. Sci. Q. 1–47.
- Liu, W., Mo, Y., Yang, Y., Ye, Z., 2015. Decision model of customer order decoupling point on multiple customer demands in logistics service supply chain. Prod. Plan. Control 26, 178–202.
- Luthans, F., Stewart, T.I., 1977. A General contingency Theory of Management. Acad. Manag. Rev. 2, 181–195. https://doi.org/10.5465/AMR.1977.4409038
- MacCarthy, B.L., 2013. An analysis of order fulfilment approaches for delivering variety and customisation. Int. J. Prod. Res. 51, 7329–7344.

- https://doi.org/10.1080/00207543.2013.852703
- Mason-Jones, R., Naylor, B., Towill, D.R., 2000. Lean, agile or leagile? Matching your supply chain to the marketplace. Int. J. Prod. Res. 38, 4061–4070. https://doi.org/10.1080/00207540050204920
- McGovern, T., Hicks, C., Earl, C.F., 1999. Modelling Supply Chain Management Processes in Engineer-to-Order Companies. Int. J. Logist. Res. Appl. 2, 147–159. https://doi.org/10.1080/13675569908901578
- Mello, M.H., Gosling, J., Naim, M.M., Strandhagen, J.O., Brett, P.O., 2017. Improving coordination in an engineer-to-order supply chain using a soft systems approach. Prod. Plan. Control. https://doi.org/10.1080/09537287.2016.1233471
- Mello, M.H., Strandhagen, J.O., Alfnes, E., 2015a. The role of coordination in avoiding project delays in an engineer-to-order supply chain. J. Manuf. Technol. Manag. https://doi.org/10.1108/JMTM-03-2013-0021
- Mello, M.H., Strandhagen, J.O., Alfnes, E., 2015b. Analyzing the factors affecting coordination in engineer-to-order supply chain. Int. J. Oper. Prod. Manag. https://doi.org/10.1108/IJOPM-12-2013-0545
- Miles, M.B., Huberman, A.M., 1994. Qualitative data analyses: An Expanded Sourcebook, 2nd ed. SAGE Publications, United States of America.
- Mintzberg, H., 1988. Generic strategies: toward a comprehensive framework. Adv. Strateg. Manag. 5, 1–67.
- Naylor, J. Ben, Naim, M., Berry, D., 1999. Leagility: integrating the lean and agile manufacturing in the total supply chain. Int. J. Prod. Econ. 62, 107–118. https://doi.org/10.1016/S0925-5273(98)00223-0
- Olhager, J., 2010. The role of the customer order decoupling point in production and supply chain management. Comput. Ind. 61, 863–868. https://doi.org/10.1016/j.compind.2010.07.011
- Olhager, J., 2003. Strategic positioning of the order penetration point. Int. J. Prod. Econ. 85, 319–329. https://doi.org/10.1016/S0925-5273(03)00119-1
- Olhager, J., Wikner, J., 2000. Production planning and control tools. Prod. Plan. Control 11, 210–222.
- Pine, B.J., 1993. Making mass customization happen: strategies for the new competitive realities. Plan. Rev. 21, 23–24.
- Rahim, A.R.A., Baksh, M.S.N., 2003. The need for a new product development framework for engineer-to-order products. Eur. J. Innov. Manag. 6, 182–196.

- https://doi.org/10.1108/14601060310486253
- Rudberg, M., Wikner, J., 2004. Mass customization in terms of the customer order decoupling point. Prod. Plan. Control 15, 445–458. https://doi.org/10.1080/0953728042000238764
- Sandrin, E., Trentin, A., Forza, C., 2018. Leveraging high-involvement practices to develop mass customization capability: A contingent configurational perspective. Int. J. Prod. Econ. 196, 335–345. https://doi.org/10.1016/j.ijpe.2017.12.005
- Schoenwitz, M., Potter, A., Gosling, J., Naim, M., 2017. Product, process and customer preference alignment in prefabricated house building. Int. J. Prod. Econ. 183, 79–90. https://doi.org/10.1016/j.ijpe.2016.10.015
- Semini, M., Gotteberg Haartveit, D.E., Alfnes, E., Arica, E., Brett, P.O., Strandhagen, J.O., 2014. Strategies for customized shipbuilding with different customer order decoupling points. Proc. Inst. Mech. Eng. Part M J. Eng. Marit. Environ. 228, 362–372. https://doi.org/10.1177/1475090213493770
- Shapiro, B.P., Rangan, V.K., Sviokla, J.J., Meisel, P. ilustraciones, 1992. Staple yourself to an order. Harvard Business Review.
- Sharman, G., 1984. The rediscovery of logistics. McKinsey Q. https://doi.org/10.1191/0267658302sr211xx
- Sinkovics, N., 2017. Pattern matching in qualitative analysis, in: The SAGE Handbook of Qualitative Business and Management Research Methods: Methods and Challenges. Sage Thousand Oaks, US, pp. 468–485.
- Sousa, R., da Silveira, G.J.C., 2018. The relationship between servitization and product customization strategies. Int. J. Oper. Prod. Manag.
- Sousa, R., Voss, C.A., 2008. Contingency Research in Operations Management Practices. J. Oper. Manag. 26, 697–713. https://doi.org/10.1016/j.jom.2008.06.001
- Stake, R.E., 1995. The Art of Case Study Research. SAGE Publications, United States of America.
- Stuart, I., McCutcheon, D., Handfield, R., McLachlin, R., Samson, D., 2002. Effective case research in operations management: A process perspective. J. Oper. Manag. https://doi.org/10.1016/S0272-6963(02)00022-0
- Sun, X.Y., Ji, P., Sun, L.Y., Wang, Y.L., 2008. Positioning multiple decoupling points in a supply network. Int. J. Prod. Econ. 113, 943–956. https://doi.org/10.1016/j.ijpe.2007.11.012
- Sylla, A., Guillon, D., Vareilles, E., Aldanondo, M., Coudert, T., Geneste, L., 2018. Configuration knowledge modeling: How to extend configuration from assemble/make to

- order towards engineer to order for the bidding process. Comput. Ind. 99, 29–41. https://doi.org/10.1016/j.compind.2018.03.019
- Thompson, J.D., 1967. Organizations in action: Social science bases of administrative theory.

  McGraw-Hill New York.
- Van Donk, D.P., 2001. Make to stock or make to order: The decoupling point in the food processing industries. Int. J. Prod. Econ. 69, 297–306. https://doi.org/10.1016/S0925-5273(00)00035-9
- Van Donk, D.P., Van Doorne, R., 2016. The impact of the customer order decoupling point on type and level of supply chain integration. Int. J. Prod. Res. 54, 2572–2584. https://doi.org/10.1080/00207543.2015.1101176
- van Nes, F., Abma, T., Jonsson, H., Deeg, D., 2010. Language differences in qualitative research: Is meaning lost in translation? Eur. J. Ageing 7, 313–316. https://doi.org/10.1007/s10433-010-0168-y
- Veldman, J., Alblas, A., 2012. Managing design variety, process variety and engineering change: a case study of two capital good firms. Res. Eng. Des. 23, 269–290.
- Venkatraman, N., 1989. The concept of fit in strategy research: Toward verbal and statistical correspondence. Acad. Manag. Rev. 14, 423–444.
- Voss, C., Johnson, M., Godsell, J., 2016. 5 Case research. Res. methods Oper. Manag. 165.
- Wemmerlöv, U., 1984. Assemble-to-order manufacturing: Implications for materials management. J. Oper. Manag. https://doi.org/10.1016/0272-6963(84)90021-4
- Wikner, J., Rudberg, M., 2005. Integrating production and engineering perspectives on the customer order decoupling point. Int. J. Oper. Prod. Manag. 25, 623–641. https://doi.org/10.1108/01443570510605072
- Willner, O., Powell, D., Gerschberger, M., Schönsleben, P., 2016. Exploring the archetypes of engineer-to-order: an empirical analysis. Int. J. Oper. Prod. Manag. 36, 242–264. https://doi.org/10.1108/IJOPM-07-2014-0339
- Wortmann, J.C., 1992. Production management systems for one-of-a-kind products. Comput. Ind. https://doi.org/10.1016/0166-3615(92)90008-B
- Wortmann, J.C., 1983. A Classification Scheme for Master Production Scheduling, in: Efficiency of Manufacturing Systems. Springer, pp. 101–109.
- Wortmann, J.C., Muntslag, D.R., Timmermans, P.J.M., 1997. Customer-driven Manufacturing. Chapman and Hall, London. https://doi.org/10.1007/978-94-009-0075-2
- Yin, R.K., 2018. Case study research and applications. Des. methods. Los Angeles.

#### **Additional sources:**

Federmacchine, 2017, "Italian machinery: solutions for all production needs", available at: http://www.federmacchine.it/en/the-sector/

Goh, 2017, "Machinery Production Market Tracker", available at:

https://technology.ihs.com/589458/machinery-production-market-tracker-q2-2017

McKinsey & Company, 2016, "How to succeed: Strategic options for European machinery", available at: <a href="https://www.mckinsey.com/~/media/McKinsey/Industries/">https://www.mckinsey.com/~/media/McKinsey/Industries/</a>

Forbes, 2018, "Major Construction Projects That Went Catastrophically Over-Budget", available at: <a href="https://www.forbes.com/sites/niallmccarthy/2018/09/28/major-construction-projects-that-went-catastrophically-over-budget-infographic/#20e9b4b46ab3">https://www.forbes.com/sites/niallmccarthy/2018/09/28/major-construction-projects-that-went-catastrophically-over-budget-infographic/#20e9b4b46ab3</a>

Forbes, 2013, "What Went Wrong at Boeing?", available at: <a href="https://www.forbes.com/sites/stevedenning/2013/01/21/what-went-wrong-at-boeing/#250f47607b1b">https://www.forbes.com/sites/stevedenning/2013/01/21/what-went-wrong-at-boeing/#250f47607b1b</a>

----- PLEASE INSERT THE APPENDIX HERE -----